
Feedback from W48

A. Akroh / F. Chapuis – FOM #23 101.12.2020

MACHINE STATUS: 4th/5 weeks of the BOOSTER COLD-CHECKOUT – 2nd & last week of Magnet Covers Consolidation (W47/48)

Hardware-Test and Cold-Checkout during Week 48:

Wide-Band Puck-up test Tomoscope, BSM and OASIS ready;

BT.BHZ10 reliability run to check it in full PPM-Mode. Pre-pulse OK in positive (to DMP/ISOLDE), missing in negative (to PS);

Issue with BTY.BVT101 External Condition solved;

Recombination Kickers External Condition behaviour issue solved;

FGC_62 firmware has been updated by TE-EPC Controls;

Reliability Run of the systems used for the Matching monitor a few missing pulse have been observed (3/11k ) investigation in progress for
TE-ABT specialists;

Dry-runs with BTTRIC and watchdogmainly working, minor issue investigation in progress;

Fast Interlock Dry-Runs: Everything finally worked! except for the BT.BHZ10.BTM (2Q power converter, when PS in access mode)  power-
converter issue, investigation in progress for TE-EPC;

BIS remaining investigation POPS-B (whenever unlocked) and RFHL (Thursday 03/12) almost everything is validated;

BE-BI: BI.DIS10 EMC investigation follow-up. Investigation still ongoing validation TBC;
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A.Akroh - F.Chapuis BE-OP Machine Markings DONE

BE-OP + M.Albert BE-OP + EN-ACE Machine final inspection and Closure

A.Akroh - F.Chapuis BE-OP Shaft Closure

M.Marcandella HSE-RP

O.Crettiez TE-MSC Magnet Covers Consolidation (All Ring) DONE

DONE

DONE

DONE

DONE

C.Bahamonde EN-SMM

Ch.Zamantzas BE-BI

A.Newborough TE-MSC + TE-EPC

A.Newborough TE-MSC + TE-EPC Ring Auxiliairy Circuits - Qstrip / BDL / Orbit Correctors and Multipoles

G.kozsian BE-RF

L.M.Coralejo TE-ABT BIr.KSW - Injection Kickers - Only W47 DONE

A.Newborough TE-MSC + TE-ABT DONE

A.Newborough TE-MSC DONE

P.Valentin EN-SMM BTM Survey DONE

BTM Magnet Lockout DONE

P.Valentin EN-SMM DONE

BI Magnet Lockout DONE

Th.Charvet EN-EL Pepraratory Visit

Equipment grounding DONE

BTY.SPT103 DONE

BTP.STP05 DONE

BTP.STP10 DONE

Potence 1T. DONE

Monorail KBK DONE

BRr.LTSA8L4

BRr.BSW4L1

BRr.BSW11L1

G.Petrika EN-CV Cooling Station Software up-date DONE

G.Petrika EN-CV BAT cooling coil Change Postponed to W48 - coil not delivred at CERN only 20/11/2020 DONE - Iced Water Connection on Monday 30/11

G.Petrika EN-CV No Air Cooling in surface building Postponed to W48 - coil not delivred at CERN only 20/11/2021 DONE - Iced Water Connection on Monday 30/12

DONE

DONE

L.Borakiewicz IT-CS DONE

TE-EPC - Timing Cables for Injection (Qstrip) DONE

TE-EPC - Ethenet QCD3 / QCD14 DONE

TE-EPC - POPSb/BIS Interlock DONE

BE-RF - FINEMET/BIS Interlock DONE

DONE

DONE

DONE

Cavity SS_13 - Potentailly Available for LL_RF Test 

Cavity SS_7 - Potentailly Available for LL_RF Test 

Cavity SS_5 - Potentailly Available for LL_RF Test 

W.Andreazza BE-BI DONE

J.Somoza TE-VSC DONE

J.Somoza TE-VSC

J.Somoza TE-VSC DONE

S.Burger BE-BI DONE

S.Burger BE-BI BEr.BTV10 - Camera Adjustment DONE

M.Marcandella HSE-RP DONE

L.M.Colarejo TE-ABT DONE

A.Newborough TE-MSC DONE

O.Crettiez TE-MSC POSTPONED to Christmas Break

W.Vigano BE-BI

CCC - Operation BE-OP Not Needed

L.de Mallac TE-EPC

CCC - Operation BE-OP

DONE

DONE

In ProgressBIr.BCT10 noise - Test with CCC  Any Time In week 48

Qstrip HWD modifications In Week 48

Validation HDW Modifications In Week 49

R.ruffieux BE-BI

BIr.BCT10 noise - shielfding Check 19/11/2020 - Morning

BIr.BCT10 noise - Visit with G.Georgiev (EN-EL) 23/11/2020 - Morning

Magnet Covers consolidation for the Trinagle - If produced in time Date no confirmed

Test of the BLM channels 30/11/2020 - 1h needed

WIC Channels Of the PSB Rings to re-check (all or partially???) ???

Survey Booster Ring and Transfer Lines 26/11/2020  (+/- 1 Day - 2h)

OASIS signal BT2.KFA20 inversed Any time in week 48

NEW INTERVENTIONS - NEW INTERVENTIONS - NEW INTERVENTIONS - NEW INTERVENTIONS - NEW INTERVENTIONS - NEW INTERVENTIONS - NEW INTERVENTIONS

BT3.QNO20 - Water leak on cooling cirduit - found Monday 16/11 19/11/2019

Feedthrough Repair - In BR20 During week 47

Leak Detection in BHZ11/151/162 During week 47

Electrical Socket Change in BTV - Surface Rack 23 & 24/11/2020 

M.E.Angoletta and 

LL-RF Team 

BE_RF (LL) Cavity tests depending: HL 

maintenance & Magnet Covers 

Consolidation progression

BWS WireScanner change in 4L1
19/11/2020

BR20 Sector Venting (from SS_2 to SS_8)

M.Haase BE-RF (HL) Maintenance Cavitiy SS_13 17/11/2020

Maintenance Cavitiy SS_7 18/11/2020

Maintenance Cavitiy SS_5 19/11/2020

26/11/2020

Ethernet Cable for BE-BI 17/11/2020 (4h Floating)

G.Georgiev EN-EL 

Floating W47/48

→  Lockout: BEr.DHZ/DVT11L1, BRr.ONO/XNO/QNO11L1

15 min in the 16/11 morning

W47
POSTPONED

POSTPONED

E.Cano EN-EL
Emergengy Exit - Crinoline - Normal Lighting 18 and 19/11/2020

Emergengy Exit - Crinoline - Emergency Lighting

16/11/2020 (2h Floating)

From 19 to 27/11/2020

→  Lockout: BTY.BVT101, BTY.QDE104, BTY.QFO108, BTY.QDE113  

→ Lockout: BT.BHZ10, BTM.BHZ10, BTM.BLM10

→  Lockout: same that Magnet Covers: BR.QFO and BR.QDE  

→  Lockout: BEr.DHZ/DVT4L1, BRr.ONO/XNO/XSK/OSK4L1

BEr.BSW - Extraction Bumpers - Only W47

BTM.QNO05 replacement →  Lockout: BTM.QNO05, BTM.BHZ10, BTM.QNO10, BT.BHZ10

from 16 to 20/11/2020

BI survey:BI.BVT.10, BI.MQN.50, BI.MQN.60 (4 niveaux), BI.BPM.40, BI.BTV.40, BI.BPM.50.

17/11/2020 (Morning)

16/11/2020 (in the day)

Vacuum Pumps Batch3 → SS:5/6/10/11/7

Vacuum Pumps Batch4 → SS:8/9 + Triangle

BRr.LTSA8L4 - Beam Scrapers

BLMs IC and FIC

Main Circuits - BR.BHZ/QFO/QDE

BRr.DES3L1 - Transverse FeedBack
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Will be confirmed during W48, 

following the intervention progresses

Magnet Covers Lockout

L

O

C

K

O

U

T

U

N

L

O

C

K

O

U

T

J.Somoza TE-VSC

Vacuum Pumps Batch1→ SS: 1/2/14/15/16

Vacuum Pumps Batch2 → SS:3/4/12/13
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A. Akroh / F. Chapuis – FOM #23 201.12.2020

Booster “Mini-Shutdown” 
(W47/48) Status

Intervention in the 
W47/W48:
The Magnets Covers 
Consolidation has been 
finished in time Friday 
27/11.

All covers are in place 
except for the external part 
of the “triangle”. Installation 
foreseen W01/02 in 2021 
Compensatory measures 
put in place for the time 
being.



Remaining issue to be fix/fixed in week 49:

 The cleanliness of the Booster (see back-up slide)

 The Booster Auxiliary circuits will be unlock out as from Tuesday 01/12;

 POPS-B Main Transformer plate forgotten  will be reassembled on Monday and Tuesday, the unlock out will follow.

Issue raised in the W48:

Change of the BR2.BWS4L1: Yet to be solved (When??? TBC);
After investigation BE-BI suspects a problem with the tank housing the BWS;

 BE-BI is studying a solution;

A. Akroh / F. Chapuis – FOM #23 301.12.2020



Thanks a lot to all of you!!
We are getting there!

Feel free to join the Daily ZOOM Chat-Room at 9:00 (L4-PSB-OP-DESK: 
https://be-dep-op.web.cern.ch/en/content/op-desk-zoom-rooms).
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https://be-dep-op.web.cern.ch/en/content/op-desk-zoom-rooms


Back-up Slide: HSE-RP + SMB + BE-OP + EN-ACE machine visit and cleaning
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